**Author’s response to reviews**

**Title:** Association of moderate-to-vigorous physical activity with neck circumference in eight Latin American countries

**Authors:**

Gerson Ferrari (gersonferrari08@yahoo.com.br)

Irina Kovalskys (ikovalskys@gmail.com)

Mauro Fisberg (mauro.fisberg@gmail.com)

Georgina Salas (georgina.gomez@ucr.ac.cr)

Attilio Rigotti (arigotti@med.puc.cl)

Lilia Sanabria (ycortes@javeriana.edu.co)

Martha García (myepez@usfq.edu.ec)

Rossina Torres (rpareja@iin.sld.pe)

Marianella Herrera-Cuenca (marianella.herrera@ucv.ve)

Ioná Zimberg (iona.zimberg@gmail.com)

Viviana Guajardo (vbgujardo@gmail.com)

Michael Pratt (michael.pratt@emory.edu)

Carlos Pires (carlosandrepires@gmail.com)

Dirceu Solé (sole.dirceu@gmail.com)

**Version:** 3 **Date:** 31 May 2019

**Author’s response to reviews:**

BMC Public Health

May 31th, 2019
Dear Natalie Pafitis and Eva Szunyogova,

Editors-in-Chief of BMC Public Health

First of all, we are grateful regarding our paper submitted to BMC Public Health by the care and attention given to the manuscript "Association of moderate-to-vigorous physical activity with neck circumference in eight Latin American countries". Our response about comments are bellow.

Editor comments

1: We note that there still remains to be some textual overlap with other sources. Please rephrase the sections of most concern using the screenshots attached.

RESPONSE: Thank you for your attention. We have correct it.

Looking forward to your response,

Yours sincerely,

Corresponding author